Author's response to reviews

Title: Feasibility and results of a randomised pilot-study of pre-discharge occupational therapy home visits

Authors:

Natasha A Lannin (nlannin@mail.usyd.edu.au)
Lindy Clemson (l.clemson@fhs.usyd.edu.au)
Annie McCluskey (a.mccluskey@usyd.edu.au)
C CHRISTINE Lin (clin2890@mail.usyd.edu.au)
Ian D Cameron (ianc@mail.usyd.edu.au)
Sarah Barras (sarah@barras.info)

Version: 2 Date: 7 January 2007

Author's response to reviews:

We have noticed a few mistakes in our manuscript and have amended accordingly. They were:

1. This sentence in the last paragraph of the Discussion "This drop-out rate would be acceptable in a larger study" was changed to "This drop-out rate would be unacceptable in a larger study"

2. A few words in Table 1 were not in alignment.